Title: A t (11; 22) (q24; q12) (EWSR1-FLI1) positive primary vaginal Ewing's sarcoma/Primitive neuroectodermal tumor (PNET) in a young girl: a case report

Authors:

Bharat Rekhi (rekhi.bharat@gmail.com)
Sajid Qureshi (sajidshafique@rediffmail.com)
Ranjan Basak (rb_tiu@yahoo.com)
Sangeeta B. Desai (sangeetabdesai@rediffmail.com)
Seema Medhi (seema.medhi@rediffmail.com)
Purna Kurkure (kurkure@vsnl.com)
Santosh Menon (santosh.menon@rediffmail.com)
Amita Maheshwari (maheshwariamita@yahoo.com)
Nirmala A. Jambhekar (najambhekar@hotmail.com)

Version: 9 Date: 2 February 2010

Author’s response to reviews:

Dear Editor Sir,

We have just added 4-5 letters of text in the abstract that was submitted in the revised manuscript but somehow did not appear in this version.

Kind regards,

Dr Bharat Rekhi